Author's response to reviews

Title: Safety in home care: A research protocol for studying medication management

Authors:

Patricia B Marck (patricia.marck@ualberta.ca)
Ariella Lang (ariella.lang@sympatico.ca)
Marilyn Macdonald (marilyn.macdonald@dal.ca)
Melissa Griffin (melissa.c.griffin@gmail.com)
Anthony Easty (tony.easty@uhn.on.ca)
Serena Corsini-Munt (serena.cm@gmail.com)

Version: 5 Date: 2 June 2010

Author's response to reviews: see over
Implementation Science
02 June 2010
Re: MS: 1719338580379301

Study protocol

Safety in Home Care: A Research Protocol for Studying Medication Management
Patricia B Marck, Ariella Lang, Marilyn Macdonald, Melissa Griffin, Anthony Easty and Serena Corsini-Munt

Dear Editor;

I have carefully re-checked this manuscript a second time to address the following requests for formatting changes to be made. I addressed # 1,2,3,4,6,7,8 but I require your help with understanding #5 please. As I go out of country in 2 days, I hope you can advise me ASAP so I can get this done before leaving.

Major revisions (author required to make these changes)

✔ 1. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. Currently the first author has a middle initial in the submission system that is not in the manuscript. Please amend this so that both author lists are identical – DONE: Patricia Marck changed to Patricia B. Marck on manuscript to match submission system.

✔ 2. Authors' contributions: The authors' contributions section mentions an author AT,
however there is not an author with these initials in the author list. Please only include authors who are present in the author list. DONE – AT should have been AE, change is made

✓ 3. References: please provide a functioning link to website reference 4. DONE

✓ 4. References: References should be cited in the text using consecutive numbers in square brackets. Currently reference 35 is out of sequence. DONE – I have re-numbered reference 35 in the text as 42 to put it in sequence; I re-ordered references 36 – 42 as 35 – 41 accordingly in the text and in the Reference List, and put old reference 35 as new reference 42 in the Reference List

* 5. Additional files: Only files cited in the main text should be attached as additional files. Please either remove the current files or cite them in order and insert an additional files section at the end of the manuscript providing the file titles and descriptions. * I do not understand this instruction. If you are referring to the proof of funding and ethics approval documents, I submitted these files at your request. Do you now wish me to remove them?

Minor revisions (I have made these changes).

✓ 6. Affiliations: Please remember to include the country in the affiliation details. DONE
7. Tracking: Please turn off track changes. DONE

8. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance. DONE

Please do not hesitate to contact me should you have any further requirements to get this manuscript to press.

Thank you,

Patricia Marck RN, PhD